Author's response to reviews

Title: Prognostic significance of proline, glutamic acid, leucine rich protein 1 (PELP1) in triple-negative breast cancer: a retrospective study on 129 cases

Authors:

Yanzhi Zhang (yeach_hmu@126.com)
Jiali Dai (daijiali999@163.com)
Keely M. McNamara (kmcnamara@patholo2.med.tohoku.ac.jp)
Bing Bai (baib_hmu@126.com)
Mumu Shi (shimumu2009@139.com)
Monica S.M. Chan (drmonicachan@hotmail.com)
Ming Liu (liuming_hmu@126.com)
Hironobu Sasano (hsasano@patholo2.med.tohoku.ac.jp)
Xiuli Wang (wangxiulinhao@sina.cn)
Xiaolei Li (xiaoleili2004@163.com)
Lijuan Liu (llj19971997@163.com)
Ying Ma (haorenmaying@126.com)
Shuwen Cao (caosw_dqytzyy@126.com)
Yanchun Xing (xingyc_lnhospital@126.com)
Baoshan Zhao (zhaobs_hmu@126.com)
Yinli Song (songyl_hmu@126.com)
Lin Wang (wangl_cmu@126.com)

Version: 6
Date: 19 August 2015

Author's response to reviews:

1 - We confirmed the approvals for this study were obtained from all the three hospitals involved and had included this information in the methods section of the manuscript.

2 - We confirmed the informed consent was written and had added this information to methods section of the manuscript.

3 - Initials had been used instead of full name in the Authors' contributions section and we confirmed the contributions of all the 17 authors were represented.